Author's response to reviews

Title: JNK signaling maintains the mesenchymal properties of multi-drug resistant human epidermoid carcinoma KB cells through snail and twist1

Authors:

Wenfu Tan (wftan@fudan.edu.cn)
Xia Zhan (xiazhan@hotmail.com)
Xiaobing Feng (xiaobing_2004@hotmail.com)
Ying Kong (kongying2010@hotmail.com)
Yi Chen (ychen@jding.dhs.org)

Version: 2 Date: 12 December 2012

Author's response to reviews: see over
Dear Sir/Madam:

This is a revised version of manuscript 2543689338677304. The only change of the revised version is that we improved the quality of written English of the manuscript by a native English speaker.

Sincerely,

Wenfu

Wenfu Tan MD. Associate Prof.
Principal Investigator
Tumor Pharmacology Section
Department of Pharmacology
School of Pharmacy, Fudan University
Research Building, Room #314
826 Zhangheng Road, Zhangjiang High-Tech. Park
Shanghai 201203, China
Tel: 86-21-51980039
Fax: 86-21-51980039
Email: wftan@fudan.edu.cn